\documentclass[11pt]{article}
\usepackage{graphicx}
\usepackage{amssymb}
\usepackage{amsmath}

\textwidth = 6.5 in \textheight = 9.5 in \oddsidemargin = 0.0 in
\evensidemargin = 0.0 in \topmargin = -0.2 in \headheight = 0.0 in
\headsep = 0.0 in
\parskip = 0.2in
\parindent = 0.0in

\newtheorem{theorem}{Theorem}
\newtheorem{corollary}[theorem]{Corollary}
\newtheorem{definition}{Definition}

\def\d{\displaystyle}
\def\b{$\bullet$}

\begin{document}
\Huge
\begin{center} Quiz 4A - MTH 1410
\end{center}

\normalsize

\textbf{Name:} \underline{\hspace{3in}}



\normalsize

\vspace{0.2in}

1) (4 points) Use logarithmic differentiation to find $\frac{dy}{dx}$ if $y=x^{\sqrt[4]{x}}$.

\newpage

2) (3 points) Find $f'(x)$ if $f(x)=\sin(\arccos x)$.  Simplify your answer.

\vspace{4in}

3) (3 points) A stone is thrown vertically upward and its height after $t$ seconds in $96t-16t^2$.  Use calculus to find the maximum height reached by the stone.







\end{document}
